Summary of May 19 Meeting to Discuss Plans and Schedule Outlook for Commissioning Cryogenics Aspects of the HINS    325 MHz Cavity Test Cave

The clean room has been moved into the cave. The roof will be put on next week.
We decided to monitor helium vessel temperature (in addition to helium bath temperature) using the sensors which will be used for the coupler in the pulsed power test. Fred Lewis will need to install the wiring for these (next week).
The helium vessel successfully passed the pressure test to 27 psi (cavity at atmospheric pressure). Timergali measured a 400 kHz frequency shift with the tuner on.  This was somewhat large and could possibly indicate that the tuner is not stiff enough; however the vessel passed the test and is approved for cooling down.
Cables from X-ray detectors need to run to the safety office on the east side of the building. Need to find out when the cable tray will be installed. If it is far off we will need to come up with a temporary solution.

Currently the 200 W amplifier is located on the rack R in the picture below. ½ in. cable follows the red plus green path. The cable is 96 ft  long. Losses are 23%, which is too much. Reducing the losses by even a few percent will speed up cavity processing time. Thus, the amplifier will be moved right outside of the cave penetration so the cable has to follow only the red path. Some extra slack within the cave will be eliminated. We can probably get down to a cable length of about 50 ft (half the losses). In addition, Dave will run 7/8’’ heliax instead of ½’’ which will reduce the losses by almost another factor of 2. (1/2’’ is 1.2 dB/100 ft, 7/8’’ is .65 dB/100 ft).
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